Author's response to reviews

Title: The growth of a culture of evidence-based obstetrics in South Africa: a qualitative case study

Authors:

Karen L Daniels Ms (karen.daniels@mrc.ac.za)
Simon A Lewin Dr (simon.lewin@nokc.no)
Practihc Policy Group (karen.daniels@mrc.ac.za)

Version: 3 Date: 28 February 2011

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Study Groups: these should be consistent between the author list in the manuscript and the submission system. Members of a group should be listed in the acknowledgements (if anywhere).

- We have now listed the Practihc Group as an author on the system and have listed the names of the group under author contributions. In the beginning of our study we agreed that the group would be listed as an author and we cannot go back on our word in this regard.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

- It has been removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- We have checked and corrected a few typographical errors, thank you.